

# LSHTM Research Online

Kuper, Hannah; Lopes Moreira, Maria Elisabeth; Barreto de Araújo, Thália Velho; Valongueiro, Sandra; Fernandes, Silke; Pinto, Marcia; Lyra, Tereza Maciel; (2019) The association of depression, anxiety, and stress with caring for a child with Congenital Zika Syndrome in Brazil; Results of a cross-sectional study. PLoS neglected tropical diseases, 13 (9). e0007768. ISSN 1935-2727 DOI: https://doi.org/10.1371/journal.pntd.0007768

Downloaded from: http://researchonline.lshtm.ac.uk/id/eprint/4654807/

DOI: https://doi.org/10.1371/journal.pntd.0007768

# **Usage Guidelines:**

Please refer to usage guidelines at https://researchonline.lshtm.ac.uk/policies.html or alternatively contact researchonline@lshtm.ac.uk.

Available under license: http://creativecommons.org/licenses/by/2.5/



# The association of depression, anxiety, and stress with caring for a child with Congenital Zika Syndrome in Brazil; Results of a cross-sectional study

Hannah Kuper 1\*, Maria Elisabeth Lopes Moreira, Thália Velho Barreto de Araújo, Sandra Valongueiro, Silke Fernandes, Marcia Pinto, Tereza Maciel Lyra, Espando es 1, Marcia Pinto, Tereza Maciel Lyra, Silke Fernandes, Marcia Pinto, Tereza Maciel Lyra, Marcia Pinto, Tereza Maciel Lyra, Silke Fernandes, Marcia Pinto, Tereza Maciel Lyra, Marcia Pinto, Tereza Maciel Lyra, Marcia Pinto, Marcia Pinto, Marcia Pinto, Marcia Pinto, Pinto, Marcia Pinto, Marcia Pinto, Pinto, Marcia Pinto, M

1 International Centre for Evidence in Disability, Clinical Research Department, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Instituto Nacional de Saúde da Mulher, da Criança e do Adolescente Fernandes Figueira, Fundação Oswaldo Cruz, Rio de Janeiro, Brasil, 3 Postgraduate Programme in Public Health, Federal University of Pernambuco, Recife, Brazil, 4 Faculty of Public Health and Policy, London School of Hygiene & Tropical Medicine, London, United Kingdom, 5 Aggeu Magalhães Institute, FIOCRUZ/PE, Recife, Brazil, 6 Faculty of Medicine, University of Pernambuco, Recife, Brazil

\* Hannah.kuper@lshtm.ac.uk



# OPEN ACCESS

Citation: Kuper H, Lopes Moreira ME, Barreto de Araújo TV, Valongueiro S, Fernandes S, Pinto M, et al. (2019) The association of depression, anxiety, and stress with caring for a child with Congenital Zika Syndrome in Brazil; Results of a cross-sectional study. PLoS Negl Trop Dis 13(9): e0007768. https://doi.org/10.1371/journal.pntd.0007768

**Editor:** Marcus VG Lacerda, Fundacao Oswaldo Cruz, BRAZIL

Received: April 16, 2019

Accepted: September 10, 2019

Published: September 30, 2019

Copyright: © 2019 Kuper et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because women may be identifiable as the number of children with CZS in these areas is relatively small, and as such, ethics was not obtained to make the data open. We are committed to collaborating with other researchers in the analysis of our data and will share data with researchers who meet the criteria for access to condiential data. Requests can be sent to the following address

# **Abstract**

# **Background**

Zika virus (ZIKV) infection in pregnancy can cause microcephaly and a wide spectrum of severe adverse outcomes, collectively called "Congenital Zika Syndrome" (CZS). Parenting a child with disabilities can have adverse mental health impacts, but these associations have not been fully explored in the context of CZS in Brazil.

#### Methodology/Principal findings

A cross-sectional study was undertaken in Recife and Rio de Janeiro, including 163 caregivers of a child with CZS (cases) and 324 caregivers with an unaffected child (comparison subjects), identified from existing studies. The primary caregiver, almost always the mother, was interviewed using a structured questionnaire to collect information on: depression, anxiety, and stress (Depression, Anxiety, and Stress Scale—DASS-21), social support (Medical Outcomes Study Social Support Scale—MOS-SSS), and socio-demographic data. Data was collected May 2017-January 2018. Ethical standards were adhered to throughout the research. A high proportion of mothers reported experiencing severe or extremely severe levels of depression (18%), anxiety (27%) and stress (36%). Mothers of children with CZS were more likely to experience symptoms of depression, anxiety andstress, compared to mothers of comparison children. These associations were more apparent among mothers living in Rio de Janeiro. These differences were reduced after adjustment for socio-economic status and social support. Among mothers of children with CZS, low social support was linked to higher levels of depression, anxiety and stress, but there was no association with socio-economic status.



(anthony.climpson-stewart@lshtm.ac.uk) for processing by the Zika Impact Data Access Committee, which consists of Professor Hannah Kuper, Dr Maria E. Moreira, Dr Tereza Lyra and Dr Gustavo Matta. In your request, please specify the variables requested and the proposed analyses.

**Funding:** This work was supported by the: Wellcome Trust & the UK's Department for International Development (205377/Z/16/Z to LCR; https://wellcome.ac.uk/); the Wellcome Trust's **Enhancing Research Activity in Epidemic Situations** programme (107779/Z/15/Z; with ER1505 & ER1601 to RAAX); the European Union's Horizon 2020 research and innovation program (https://ec. europa.eu/programmes/horizon2020/) under ZikaPLAN grant agreement No. 734584 (https:// zikaplan.tghn.org/); the Medical Research Council on behalf of the Newton Fund and the Wellcome Trust (MC\_PC\_15088 to LCR; https://mrc.ukri.org/ ); Secretaria de Vigilância em Saúde/ Ministério da Saúde de Brasil Resposta à Emergência em Saúde Pública - Zika vírus e Microcefalia (837058/2016 to RAAX); the Conselho Nacional de Desenvolvimento Científico e Tecnológico, the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior (http://www.capes.gov.br/); & the Departamento de Ciência e Tecnologia (http://rebrats.saude.gov. br/noticias/59DECIT) for Prevenção e Combate ao vírus Zika I (440839/2016-5 to RAAX). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

# Conclusions/Significance

Depression, anxiety and stress were very common among mothers of young children in Brazil, regardless of whether they were parenting a child with disabilities. Mothers of children with CZS may be particularly vulnerable to poor mental health, and this association may be buffered through better social support.

# **Author summary**

The 2015 Zika epidemic in Brazil gained international attention with the birth of thousands of babies with severe adverse outcomes, collectively called "Congenital Zika Syndrome" (CZS). Parenting a child with disabilities can be extremely stressful, and in other settings is linked to depression, anxiety and other mental health conditions. However, these associations have not been fully explored in the context of CZS in Brazil. We conducted a cross-sectional study in Rio de Janeiro and Recife, two cities in Brazil. We identified 163 caregivers (mostly the mother) of a child with CZS, and 324 caregivers of children of similar ages, but without obvious disabilities. The caregivers were asked a series of questions to gauge whether they were experiencing depression, anxiety or stress. We found that a high proportion of the mothers reported experiencing severe or extremely severe levels of depression (18%), anxiety (27%) and stress (36%). Levels of depression and anxiety were even higher for mothers of children with CZS compared to mothers of unaffected children, especially for those mothers with low social support and those living in Rio de Janeiro.

#### Introduction

Brazil hit international headlines in late 2015 with the birth of thousands of babies with microcephaly, which was soon linked to congenital infection with the Zika Virus (ZIKV). It was quickly apparent that babies born after congenital ZIKV infection experience a range of severe conditions beyond microcephaly alone, which is now collectively called "Congenital Zika Syndrome" (CZS). [1] Phenotypic characterization is still ongoing, but to date CZS is indicated by: microcephaly, other patterns of brain damage (including subcortical calcifications), damage to the back of the eye, congenital contractures, and hypertonia restricting body movement soon after birth. Children with CZS therefore have multiple and broad-ranging impairments, and as a result have complex care needs, which lie mostly with parents, in particular the mother. By early 2018, 3,149 cases of CZS were registered in Brazil, with many more potential cases not officially confirmed. [2]

Evidence is growing that carers of children with disabilities are more likely to experience depression, anxiety and stress, [3–5] and this may make it more difficult to meet the care needs of their child. Mental health impacts arise through different potential pathways. It may be distressing and overwhelming to meet the complex needs of a child with disabilities, which will continue for the lifetime of the child. Prevalence of depression and anxiety may therefore be particularly high among the carers of children with greater functional difficulties. [3, 4] Parents of children with disabilities may also experience negative attitudes and discrimination, and consequent social isolation, including marital breakdown, [6, 7] which can contribute to poor mental health. Caring for a severely disabled child often has cost implications, both direct

costs (e.g. treatment) and indirect costs (e.g. foregone wages as the mother does not return to work), [8] creating financial problems, [7] and thereby increasing the risk of depression, anxiety and stress. [9] On the positive side, good social support and access to adequate resources may buffer the negative mental health impacts for carers.[10]

Carers of children with CZS may be particularly vulnerable to poor mental health. [11] These children experience severe impairments, with high healthcare needs. Affected children may be highly irritable and difficult to care for—as one mother described: "For every ten minutes of sleep, she cries for an hour".[12] Another source of distress in relation to CZS is the uncertainty about the long-term trajectory of the condition, given its newness, and a lack of specialized professional knowledge. Families of children with CZS are also on average poorer and therefore may be less resilient to these challenges. [13] However, limited evidence is available on the impacts of CZS on the mental health of carers, despite the strong case for why a link is likely. One small study, conducted in Sergipe State in Brazil compared levels of anxiety and depression in the first 24 hours after birth between 9 mothers of newborns with microcephaly and twenty mothers with healthy newborns. [14] Mothers of babies with microcephaly had lower scores in the psychological domain of quality of life than those with healthy newborns, but there was no difference in anxiety (which was high in both groups) or depression. After ten months of follow-up, the mothers of children with CZS reported high levels of anxiety and low quality of life.[15] However, these results must be interpreted with caution due to the very small sample size, and other relevant studies are lacking.

In order to address these gaps in the literature, the current study aimed to explore the association of having a child affected by CZS with depression, anxiety, and stress, and to assess whether these relationships are buffered by social support and socio-economic status.

#### Methods

# Overview and setting

The methods of the study have been described in full elsewhere. [16] Briefly, a cross-sectional study was undertaken to explore the differences between caregivers with a child affected by CZS to those of typically developing children, in terms of social and economic indicators. Two contrasting sites were selected. The first was metropolitan areas in Recife, in the State of Pernambuco in Northeast Brazil, which had a high number of suspected and confirmed cases of CZS. [2] The second site was Rio de Janeiro City, where symptomatic ZIKV was less prevalent and reports of CZS far lower.

#### Sample size

We aimed to recruit 100 cases of children with CZS and 100 comparison subjects without CZS per setting, which would provide the power to detect an OR of 2.6 in each site for the association between depression and CZS, assuming 95% confidence, 80% power and a prevalence of depression of 15% in unaffected caregivers. [17] Across the two samples (i.e. 200 cases and 200 comparison subjects), the sample size would be adequate to detect an OR of 2.05 for the same association.

#### Recruitment of cases and comparison subjects

In Recife, the source of most of the cases and all the comparison subjects was an existing case-control study, initiated in January 2016 to identify causes of CZS. [18] Cases were children born with microcephaly (head circumferences < 2 SD than the mean) in eight public maternity hospitals in Recife. Controls were children born in the same hospitals, but without

microcephaly and without apparent neurological or other health problems (determined from transfontanellar ultrasonography, and through physical examination by the study neonatologist), with both examinations performed soon after birth. Controls were matched to cases on the basis of expected date of delivery and place of mother's residence (by Health Region). At follow-up, caregivers of controls were asked whether there were any developmental delays (using the Denver test) and if the response was positive, the control child was excluded from the study and referred for further investigation. Additional cases with CZS were identified from an ongoing "cohort of children". These children were identified as potentially having CZS from those born to a cohort of pregnant women who presented with a rash (a common symptom of ZIKV infection), and from outpatient clinics of children with CZS. Suspect cases were examined by a pool of specialists to confirm CZS. At follow-up, the participants were are classified as "cases" and "comparison subjects" (rather than cases and controls), to reflect the fact that we were assessing cross-sectional association of mental health conditions and CZS, rather than identifying exposures which may be aetiologically linked to CZS.

In Rio de Janeiro, the source of the cases and comparison subjects was the Vertical Exposure to Zika Virus and Its Consequences for Child Neurodevelopment: Cohort Study in Fiocruz/IFF (ClinicalTrials.gov Identifier: NCT03255369). Cases were children born to mothers known to be ZIKV positive, who had microcephaly. Comparison subjects were born to mothers without a history of symptoms and without developmental delay, as shown by: 1) a composite Bayley Score  $\geq$ 85 conducted between 6 and 36 months following the recommended guidelines and/or 2) assessment by two paediatricians based on the child's medical records. [19]

#### Data collection

The primary caregiver, usually the mother, was interviewed using a structured questionnaire. The Depression, Anxiety, and Stress Scale (DASS-21) was used to assess symptoms of depression, anxiety, and stress among caregivers. [20] It is a 21-item questionnaire with a four-point (0–3) answer scale that focuses on the extent participants had experienced certain symptoms over the previous week. Items are arranged into three subscales (depression, anxiety and stress), with seven items for each subscale. DASS-21 is a reliable tool to assess psychological distress and has been adapted and validated for Brazilian Portuguese. [21] The Medical Outcomes Study Social Support Scale (MOS-SSS) was used to measure social support. It is a 19-item questionnaire with each item scored on a Likert scale of 1 to 5, which includes five scales covering different aspects of social support (affection, positive social interaction, emotional, informational, and material). It has also been validated for Brazilian Portuguese. [22] Data was also collected on: the parents' socio-demographic characteristics (e.g. asset ownership, parental marital status).

Data collection was undertaken between May 2017 and January, 2018. In Recife, the caregivers were interviewed in their homes, at the Primary Health Centre or occasionally, in their workplace. In Rio de Janeiro the interview was undertaken in person at attendance at The Fernandes Figueira Institute (IFF). In Recife, four quantitative interviewers were included (all female) and three were included in Rio de Janeiro (two female one male), all had a degree in social science or public health.

#### Quantitative data analysis

Variables were created from the standardized questionnaires. For DASS-21, sub-scales were calculated for Depression, Anxiety, and Stress Scale with each subclass's score equal to the sum of seven corresponding questions. The sum scores were multiplied by 2 in order to match the



original scale score in DASS-42 so that each subscale score ranges from 0 to 42. [20] Categories were created for:

- Depression (normal: <9; mild: 10–13, moderate: 14–20, severe: 21–27, extremely severe: >28).
- Anxiety (normal: <7; mild: 8–9, moderate: 10–14, severe: 15–19, extremely severe: >20).
- Stress (normal: <14; mild: 15–18, moderate: 19–25, severe: 26–33, extremely severe: >= 34).

For MOS-SSS, an overall social support index was calculated, ranging from 0 to 100, with higher score indicating better availability of social support. In addition, four separate social support functional subscales were generated measuring: 1) emotional/informational social support, 2) tangible social support, 3) affectionate social support, and 4) positive social interaction.

Socio-demographic characteristics were regrouped into categories, to reflect the distribution of the data among the participants (groupings shown in Table 1). The Social-economic strata were estimated following the Brazilian Criteria, based on household assets from a predefined list, head of household education as well as the household's access to piped water and a paved street. The household income estimates in each strata are approximations based on the 2013 National Household Sample Survey. [23]

A chi-square test was used to compare the socio-demographic characteristics of cases and comparison subjects, to identify potential confounders of the association of CZS with mental health outcomes. The mean depression, anxiety and stress levels were compared for carers of children with CZS and comparison subjects using a Wilcoxon rank sum test.

Multivariable logistic regression analyses were undertaken using Stata (version 15) to compare the odds of study outcomes (i.e. maternal depression, anxiety and stress) among carers of children with CZS to those of unaffected children, adjusted for age, socio-economic status (SES) variables and location (Rio de Janeiro/Recife). We tested associations for effect-modification by, in turn, location, social support and economic status. There were few missing data, and analyses were conducted for participants where complete data for relevant indicators were available.

#### **Ethics statement**

Ethical approval for the full study was received from LSHTM and the Fiocruz ethics committee (CAAE 60682516.2.1001.5269). The original case-control study in Recife was approved by the Research Ethics Committees of the Pan American Health Organization (PAHO-2015-12-0075) and Fiocruz Pernambuco (CAAE: 51849215.9.0000.5190) and the Cohort study in Rio de Janeiro was approved by the IFF Ethics Committee (CAAE 52675616.0.0000.5192). The "Cohort of children" in Recife was approved by the ethics Committee of the Oswaldo Cruz University Hospital, University of Pernambuco (CAAE: 52803316.8.0000.5192). All interviewees were adults and all provided written informed consent.

#### Results

The study included 81 cases and 112 comparison subjects from Recife, and 82 cases and 155 comparison subjects from Rio de Janeiro (Table 1). Among those interviewed, 97.5% were the mother, 1.4% the father and 1.0% the grandmother, and so the carer is referred to as the "mother" throughout for convenience. In both locations, mothers of cases were significantly younger than mothers of comparison subjects. Furthermore, in Recife the case children were older than the comparison children, whereas in Rio de Janeiro the reverse was true. In Rio de



Table 1. Baseline characteristics of mothers with a child with microcephaly and comparison mothers.

		<b>Recife</b> (n = 193)	Rio de Janeiro (n = 237)			
	Cases (n = 81)	Comparison subjects (n = 112)	X <sup>2</sup> p-value	Cases (n = 82)	Comparison subjects (n = 155)	X <sup>2</sup> p- value
Age group mother						
<20 years	13%	10%	0.27	11%	4%	0.008
20-29 years	48%	54%		53%	41%	
30-39 years	33%	34%		34%	44%	
40+years	8%	2%		3%	11%	
Marital status						
Married	33%	32%	0.85	28%	45%	0.03
Cohabiting	38%	36%		43%	37%	
Other	28%	32%		29%	19%	
Age of child						
<12 months	10%	27%	< 0.001	50%	28%	0.004
12 months-<18months	14%	65%		35%	54%	
18 months+	77%	8%		15%	18%	
Gender of child						
Male	43%	55%	0.10	50%	47%	0.67
Female	57%	45%		50%	53%	
Education mother						
None-Junior high completed	57%	49%	0.43	37%	25%	0.01
High school completed	43%	50%		59%	57%	
University degree completed	0%	1%		5%	17%	
Economic strata (reference average income per strata)						
A-B (R\$ 4427–20273)	4%	4%	0.48	17%	32%	0.03
C1 (R\$ 2409)	16%	11%		23%	27%	
C2 (R\$1446)	30%	39%		34%	27%	
D-E (R\$ 640)	51%	46%		26%	14%	
Number of people living in household (incl. child)						
≤4	40%	32%	0.57	43%	40%	0.49
4	27%	31%		27%	34%	
≥5	33%	37%		30%	26%	

Janeiro, the case mother was less likely to be living in a union (71% versus 81% for comparison subjects), was less likely to be educated beyond the primary level (64% versus 74% for comparison subjects), and was in a lower economic stratum as compared to the comparison subjects. These differences were not apparent in the Recife sample, and in both locations the cases and comparison subjects were similar in terms of the number of people living in the household and gender of the child.

Median depression (p = 0.01), anxiety (p = 0.03) and stress scores (p = 0.05) were higher among cases than comparison subjects (Table 2). After stratification by location it was apparent that in Rio de Janeiro, scores were significantly higher for cases than comparison subjects for depression (p = 0.02), anxiety (p = 0.006) and stress (p = 0.02), while no differences were apparent in Recife between cases in comparison subjects.

Overall, a high proportion of mothers (across cases and comparison subjects) reported experiencing severe or extremely severe levels of depression (18%), anxiety (27%) and stress



Table 2. Comparison of depression, anxiety and stress scores among cases and comparison subjects, by location.

	Location		N	Median	Inter-quartile range	Wilcoxon rank-sum p-value
Depression score	Recife	Cases	81	8	4-20	0.28
		Comparisons	112	6	2–18	
	Rio de Janeiro	Cases	82	10	2–20	0.02
		Comparisons	155	6	2–18	
	All	Cases	163	10	2–20	0.01
		Comparisons	267	6	2–16	
Anxiety score	Recife	Cases	81	8	2-18	0.92
,		Comparisons	112	6	2-18	
	Rio de Janeiro	Cases	82	8	4–18	0.006
		Comparisons	155	4	0-12	
	All	Cases	163	8	2-18	0.03
		Comparisons	267	6	0-16	
Stress score	Recife	Cases	81	16	10-32	0.75
		Comparisons	112	18	8-30	
	Rio de Janeiro	Cases	82	22	12-34	0.02
		Comparisons	155	18	8–26	
	All	Cases	163	20	10-34	0.05
		Comparisons	267	18	8-28	

(36%) (Table 3). Unadjusted analyses showed that mothers of children with CZS showed higher levels of "any" depression (OR = 1.5, 95% CI = 1.0-2.2) or "any" anxiety (1.4, 1.0-2.1) compared to comparison subjects. However, these associations were not apparent after adjustment for socio-economic status and social support. There did not appear to be a trend in this

Table 3. Depression, anxiety and stress in mothers with a child with CZS, and comparison subjects.

	Cases (n = 163)	Comparison subjects (n = 267)	Unadjusted OR (95% CI)	OR adjusted for maternal age and child's age (95% CI)	OR adjusted for maternal age, child's age and socio-economic status (95% CI)	OR adjusted for maternal age, child's age and socio-economic status and overall social support (95% CI)
	%	%				
Depression						
Normal	49%	58%	1.0	1.0	1.0	1.0
Mild/moderate	30%	25%	1.4 (0.9-2.3)	1.5 (0.9-2.6)	1.5 (0.9–2.5)	1.3 (0.8–2.3)
Severe/ extremely severe	21%	16%	1.5 (0.9–2.5)	1.5 (0.8-2.7)	1.4 (0.8–2.5)	1.3 (0.7–2.4)
Any depression (≥mild)	51%	42%	1.5 (1.0-2.2)	1.5 (1.0-2.3)	1.4 (0.9-2.2)	1.3 (0.8–2.0)
Anxiety						
Normal	47%	56%	1.0	1.0	1.0	1.0
Mild/moderate	22%	19%	1.4 (0.8-2.3)	1.1 (0.6-2.0)	1.2 (0.6–2.1)	1.1 (0.6–1.9)
Severe/ extremely severe	31%	25%	1.5 (0.9–2.4)	1.5 (0.9-2.4)	1.4 (0.8–2.3)	1.3 (0.7–2.2)
Any anxiety (≥mild)	53%	44%	1.4 (1.0-2.1)	1.3 (0.8-2.0)	1.2 (0.8–1.9)	1.1 (0.7–1.8)
Stress						
Normal	40%	43%	1.0	1.0	1.0	1.0
Mild/moderate	19%	24%	0.9 (0.5-1.4)	1.1 (0.6–2.0)	1.2 (0.6-2.1)	1.1 (0.6-2.1)
Severe/ extremely severe	41%	32%	1.4 (0.9-2.2)	1.5 (0.9-2.4)	1.4 (0.9-2.4)	1.3 (0.7-2.1)
Any stress (≥mild)	60%	57%	1.2 (0.8-1.7)	1.3 (0.8-2.0)	1.3 (0.8-2.0)	1.2 (0.7-1.9)

https://doi.org/10.1371/journal.pntd.0007768.t003



Table 4. Association of social support with depression, anxiety and stress among mothers of child with CZS.

Component of social support	Any depression (N = 83) versus none (N = 80) OR (95%CI)	Any anxiety (N = 87) versus none (N = 76) OR (95%CI)	Any stress (N = 98) versus none (N = 65) OR (95%CI)
Emotional/informational support			
All the time	1.0	1.0	1.0
Most of the time	2.3 (1.0-5.5)	2.5 (1.1-6.0)	2.4 (1.0-5.8)
Some, a little, none of the time	6.4 (2.9–14.0)	2.9 (1.4–6.0)	4.2 (1.9-9.1)
Tangible support			
All the time	1.0	1.0	1.0
Most of the time	1.5 (0.7–3.2)	1.3 (0.6–3.0)	1.1 (0.5–2.5)
Some, a little, none of the time	2.6 (1.3–5.5)	1.6 (0.8–3.2)	1.2 (0.6–2.5)
Affectionate support			
All the time	1.0	1.0	1.0
Most of the time	2.0 (0.9–4.5)	1.9 (0.8-4.3)	1.6 (0.7–3.7)
Some, a little, none of the time	3.1 (1.2–7.7)	1.8 (0.8–4.4)	1.9 (0.8–4.7)
Positive social interaction			
All the time	1.0	1.0	1.0
Most of the time	3.1 (1.4–7.1)	1.6 (0.7–3.7)	2.0 (0.9–4.4)
Some, a little, none of the time	6.2 (2.8–13.5)	5.6 (2.5–12.4)	5.3 (2.3–12.2)
Overall social support			
All the time	1.0	1.0	1.0
Most of the time	2.6 (1.2–5.7)	1.6 (0.7–3.3)	1.7 (0.8–3.6)
Some, a little, none of the time	6.9 (2.9–16.4)	3.1 (1.4-6.9)	3.6 (1.5-8.4)

association by severity of depression or anxiety. No differences in stress levels were identified between cases and comparison subjects.

The association between depression, anxiety and stress (in turn) and components of social support were investigated among the mothers of children with CZS (<u>Table 4</u>). Mothers of children with CZS were more likely to experience symptoms of depression if they had low levels of social support, and this was consistent across all domains of social support. For anxiety and stress, this pattern was apparent in relation to overall social support, positive social interaction, and emotional/informational support, but not for the other domains of social support.

The association between parenting a child with CZS and depression, anxiety and stress (in turn) was stratified by socio-economic status, social support and location (Table 5). No significant effect modification was detected for any of these comparisons Parents of children with CZS living in Rio de Janeiro appeared more likely to experience depression, anxiety and stress than parents of children with CZS living in Recife, but the p-value for interaction did not reach statistical significance.

#### **Discussion**

This cross-sectional study compared the mental health of mothers of children with CZS to those with developmentally typical children in Brazil. The study found that a high proportion of mothers reported experiencing severe or extremely severe levels of depression, anxiety and stress, even among mothers of children without apparent disabilities. Mothers of children with CZS were more likely to experience symptoms of depression and anxiety, but not stress, and these differences disappeared after adjustment for socio-economic status and social support. Among mothers of children with CZS, low social support was linked to higher levels of



Table 5. The association of depression, anxiety and stress with parenting a child with CZS, stratified by SES, social support and location.

	Socio-economic status				Location				
	Low SES (C2-E) (n = 274)	High SES (A-C1) (n = 156)	P for interaction	Low social support (≤some of the time) (n = 73)	High social support (≥most of the time) (n = 355)	P for interaction	Recife (n = 193)	Rio de Janeiro (n = 237)	P for interaction
	OR (95% CI)	OR (95% CI)		OR (95% CI)	OR (95% CI)		OR (95% CI)	OR (95% CI)	
Depression									
None	1.0	1.0		1.0	1.0		1.0	1.0	
Any	1.5 (0.9-2.4)	1.2 (0.6-2.4)	0.60	1.9 (0.7-5.3)	1.1 (0.7-1.8)	0.40	1.1 (0.6–1.9)	1.9 (1.1-3.3)	0.14
Anxiety									
None	1.0	1.0		1.0	1.0		1.0	1.0	
Any	1.5 (0.9-2.4)	1.3 (0.7-2.6)	0.82	1.3 (0.5-3.4)	1.3 (0.8-2.0)	0.99	1.2 (0.7-2.2)	1.6 (0.9-2.8)	0.51
Stress									
None	1.0	1.0		1.0	1.0		1.0	1.0	
Any	1.1 (0.7-1.8)	1.2 (0.6-2.3)	0.93	1.4 (0.5-4.0)	1.0 (0.6-1.5)	0.54	0.9 (0.5-1.6)	1.5 (0.9–2.6)	0.22

depression, anxiety and stress. After stratification by location, anxiety and depression were associated with parenting a child with CZS in Rio de Janeiro, but not in Recife.

Previous studies have consistently demonstrated an association between parenting a child with neurodevelopmental disabilities and the experiences of depression, anxiety and stress.[3– 5, 9] We can speculate on the reason for the disparity with the results of the current study, where these associations were less strong and consistent. ZIKV was reported about widely in the media in Brazil, and so there was a high awareness of the condition, and potentially less stigma experienced by parents than occurs for other types of childhood disability. The Brazilian government committed to the support of children with CZS and their families, and consequently many families may be able to access to health services and social benefits, reducing financial stress. There are also some social support structures in Brazil for parents of CZS, ranging from informal Whatsapp groups to established NGOs, which may help to buffer the negative mental health impacts of caring for a child with disabilities. Furthermore, poor mental health appeared to be extremely common among the comparison subjects, making it more difficult to detect a difference with cases, perhaps because of the pervasive social issues experienced in Brazil currently. The negative mental health impacts of having a child with CZS was more pronounced in parents living in Rio de Janeiro than in Recife. A potential explanation may be that the comparison mothers in Rio de Janeiro were generally older, more likely to be living in a union, and had higher levels of education and socio-economic status than the mothers in the other groups, which may influence the lower levels of depression and anxiety. In Recife, the cases and comparison subjects were matched by location, and as women living in poverty and vulnerable situations are more likely to experience mental health concerns, differences between cases and comparison subjects would have been reduced.

The mental health of carers of children with disabilities remains an important concern. However, in this context the high prevalence of depression, anxiety and stress among mothers of unaffected children was disturbing and needs urgent attention. Depression, anxiety and stress reduce the quality of life of those affected and are also linked to poorer physical health and social functioning. [24, 25] Parental depression can also result in negative parenting behaviours and reduced interaction with the child, so that the child may not receive the stimulation needed to thrive, [24, 25] which is particularly of concern in the case of children with disabilities, who are already facing challenges in their development. It is therefore important to identify the mental health needs of parents of children with disabilities, so that appropriate

interventions can be implemented. A recent Cochrane review reported that psychological therapies improved parental mental health for parents of children with cancer post- treatment, [26] although evidence was lacking of positive impacts for parents of children with other chronic conditions. Improving social support may also help to buffer the negative impacts of caring on mental health, for instance, through establishing parent support groups, offering relationship counselling, or developing informal support networks (e.g. Whatsapp groups). [10, 11, 27] Improving mental health and social support of carers, may also help to improve adaptive parenting behavior, [26] and thereby benefit the child. Currently, more research is needed on which interventions work to improve mental health and social support of carers, particularly in low resource settings.

There are important strengths and limitations of the study, which must be taken into account when considering the findings. This was a relatively large study, conducted across two contrasting sites, and including comprehensive questionnaires using validated scales for depression, anxiety, stress and social support. This study also fills an important gap, as currently robust quantitative data on the mental health impacts of carers of children with CZS is lacking. In terms of limitations, clinical diagnosis of mental health conditions in the parents was lacking, and so we relied on screening questionnaires for assessing symptoms. The number of cases recruited was smaller than planned (though the number of comparison subjectswas higher), and the association between depression and CZS was weaker than anticipated, and so some of the analyses may have been under-powered. In particular, we were under-powered to identify effect modifiers.

#### **Conclusions**

This study found that depression, anxiety and stress were very common among mothers of young children in Brazil. Mothers of children with CZS may be particularly vulnerable to poor mental health, and this impact may be buffered by improving social support.

### Supporting information

**S1** Checklist. Strobe checklist. (DOC)

#### **Acknowledgments**

We are grateful to all the participants who contributed their time to our project. We would also like to acknowledge our colleagues who collaborated on the Social and Economic Impact of Zika Project.

## **Author Contributions**

Conceptualization: Hannah Kuper, Maria Elisabeth Lopes Moreira, Thália Velho Barreto de Araújo, Sandra Valongueiro, Tereza Maciel Lyra.

Data curation: Thália Velho Barreto de Araújo, Sandra Valongueiro.

Formal analysis: Silke Fernandes, Tereza Maciel Lyra.

**Funding acquisition:** Hannah Kuper, Maria Elisabeth Lopes Moreira, Thália Velho Barreto de Araújo, Sandra Valongueiro, Tereza Maciel Lyra.

**Investigation:** Maria Elisabeth Lopes Moreira, Thália Velho Barreto de Araújo, Sandra Valongueiro, Tereza Maciel Lyra.



**Methodology:** Hannah Kuper, Thália Velho Barreto de Araújo, Sandra Valongueiro, Silke Fernandes, Marcia Pinto, Tereza Maciel Lyra.

**Project administration:** Maria Elisabeth Lopes Moreira, Thália Velho Barreto de Araújo, Sandra Valongueiro, Marcia Pinto, Tereza Maciel Lyra.

**Supervision:** Hannah Kuper, Maria Elisabeth Lopes Moreira, Thália Velho Barreto de Araújo, Sandra Valongueiro, Marcia Pinto.

Writing - original draft: Hannah Kuper, Silke Fernandes.

Writing – review & editing: Hannah Kuper, Maria Elisabeth Lopes Moreira, Thália Velho Barreto de Araújo, Sandra Valongueiro, Silke Fernandes, Marcia Pinto, Tereza Maciel Lyra.

#### References

- Moore CA, Staples JE, Dobyns WB, Pessoa A, Ventura CV, Fonseca EB, et al. Characterizing the Pattern of Anomalies in Congenital Zika Syndrome for Pediatric Clinicians. JAMA Pediatr. 2017; 171(3):288–95. Epub 2016/11/05. https://doi.org/10.1001/jamapediatrics.2016.3982 PMID: 27812690.
- Monitoramento integrado de alterações no crescimento e desenvolvimento relacionadas à infecção pelo vírus Zika e outras etiologias infecciosas, até a Semana Epidemiológica 20/2018. [Internet]. 2018.
- Singer GH. Meta-analysis of comparative studies of depression in mothers of children with and without developmental disabilities. American journal of mental retardation: AJMR. 2006; 111(3):155–69. Epub 2006/04/07. PMID: 16597183.
- Yilmaz H, Erkin G, Nalbant L. Depression and anxiety levels in mothers of children with cerebral palsy: a controlled study. Eur J Phys Rehabil Med. 2013; 49(6):823–7. Epub 2013/10/10. PMID: 24104700.
- Pinquart M. Parenting stress in caregivers of children with chronic physical condition-A meta-analysis. Stress and health: journal of the International Society for the Investigation of Stress. 2018; 34(2):197–207. Epub 2017/08/24. https://doi.org/10.1002/smi.2780 PMID: 28834111.
- Tilahun D, Hanlon C, Fekadu A, Tekola B, Baheretibeb Y, Hoekstra RA. Stigma, explanatory models and unmet needs of caregivers of children with developmental disorders in a low-income African country: a cross-sectional facility-based survey. BMC Health Serv Res. 2016; 16:152. Epub 2016/04/28. https://doi.org/10.1186/s12913-016-1383-9 PMID: 27117326.
- DiGiacomo M, Green A, Delaney P, Delaney J, Patradoon-Ho P, Davidson PM, et al. Experiences and needs of carers of Aboriginal children with a disability: a qualitative study. BMC Fam Pract. 2017; 18(1):96. Epub 2017/12/01. https://doi.org/10.1186/s12875-017-0668-3 PMID: 29187141.
- Stabile M, Allin S. The economic costs of childhood disability. Future Child. 2012; 22(1):65–96. Epub 2012/05/04. PMID: 22550686.
- Unsal-Delialioglu S, Kaya K, Ozel S, Gorgulu G. Depression in mothers of children with cerebral palsy and related factors in Turkey: a controlled study. International journal of rehabilitation research Internationale Zeitschrift fur Rehabilitationsforschung Revue internationale de recherches de readaptation. 2009; 32(3):199–204. Epub 2009/03/27. https://doi.org/10.1097/MRR.0b013e32832607b6 PMID: 19322102.
- Lindo EJ, Kliemann KR, Combes BH, Frank J. Managing Stress Levels of Parents of Children with Developmental Disabilities: A Meta-Analytic Review of Interventions. Family Relations. 2016; 65 (1):207–24. https://doi.org/10.1111/fare.12185
- Bailey DB Jr., Ventura LO. The Likely Impact of Congenital Zika Syndrome on Families: Considerations for Family Supports and Services. Pediatrics. 2018; 141(Suppl 2):S180–s7. Epub 2018/02/14. https:// doi.org/10.1542/peds.2017-2038G PMID: 29437051 conflicts of interest to disclose.
- 12. Diniz D. Zika virus and women. Cad Saude Publica. 2016; 32(5):e00046316. PMID: 27192024.
- Souza WV, Albuquerque M, Vazquez E, Bezerra LCA, Mendes A, Lyra TM, et al. Microcephaly epidemic related to the Zika virus and living conditions in Recife, Northeast Brazil. BMC public health. 2018; 18(1):130. Epub 2018/01/14. https://doi.org/10.1186/s12889-018-5039-z PMID: 29329574.
- Dos Santos Oliveira SJG, de Melo ES, Reinheimer DM, Gurgel RQ, Santos VS, Martins-Filho PRS. Anxiety, depression, and quality of life in mothers of newborns with microcephaly and presumed congenital Zika virus infection. Arch Womens Ment Health. 2016; 19(6):1149–51. Epub 2016/08/20. <a href="https://doi.org/10.1007/s00737-016-0654-0">https://doi.org/10.1007/s00737-016-0654-0</a> PMID: 27538400.
- Dos Santos Oliveira SJG, Dos Reis CL, Cipolotti R, Gurgel RQ, Santos VS, Martins-Filho PRS. Anxiety, depression, and quality of life in mothers of newborns with microcephaly and presumed congenital Zika



- virus infection: a follow-up study during the first year after birth. Arch Womens Ment Health. 2017; 20(3):473–5. Epub 2017/04/22. https://doi.org/10.1007/s00737-017-0724-y PMID: 28429099.
- 16. Kuper H, Lyra T, Moreira M, de Albuquerque M, de Araújo T, Fernandes S, et al. Social and economic impacts of congenital Zika syndrome in Brazil: Study protocol and rationale for a mixed-methods study [version 1; referees: 1 approved]. Wellcome Open Research. 2018; 3(127). https://doi.org/10.12688/wellcomeopenres.14838.1
- 17. Siqueira Barcelos R, da Silva Dos Santos I, Matijasevich A, Anselmi L, Barros FC. Maternal depression is associated with injuries in children aged 2–4 years: the Pelotas 2004 Birth Cohort. Inj Prev. 2018. Epub 2018/02/28. https://doi.org/10.1136/injuryprev-2017-042641 PMID: 29483241.
- 18. de Araujo TVB, Ximenes RAA, Miranda-Filho DB, Souza WV, Montarroyos UR, de Melo APL, et al. Association between microcephaly, Zika virus infection, and other risk factors in Brazil: final report of a case-control study. Lancet Infect Dis. 2018; 18(3):328–36. Epub 2017/12/16. https://doi.org/10.1016/S1473-3099(17)30727-2 PMID: 29242091.
- Brasil P, Pereira JP Jr., Moreira ME, Ribeiro Nogueira RM, Damasceno L, Wakimoto M, et al. Zika Virus Infection in Pregnant Women in Rio de Janeiro. N Engl J Med. 2016; 375(24):2321–34. Epub 2016/03/ 05. https://doi.org/10.1056/NEJMoa1602412 PMID: 26943629.
- Lovibond S, Lovibond P. Manual for the Depression Anxiety Stress Scales. Sydney: Psychology Foundation, 1995.
- Vignola RC, Tucci AM. Adaptation and validation of the depression, anxiety and stress scale (DASS) to Brazilian Portuguese. Journal of affective disorders. 2014; 155:104–9. Epub 2013/11/19. https://doi. org/10.1016/j.jad.2013.10.031 PMID: 24238871.
- Soares A, Biasoli I, Scheliga A, Baptista RL, Brabo EP, Morais JC, et al. Validation of the Brazilian Portuguese version of the Medical Outcomes Study-Social Support Survey in Hodgkin's lymphoma survivors. Support Care Cancer. 2012; 20(8):1895–900. Epub 2011/10/07. https://doi.org/10.1007/s00520-011-1292-8 PMID: 21976036.
- Pesquisa ABdEd. Criterio de classificacao economica Brasil 2014. Available from: <a href="http://www.abep.org/">http://www.abep.org/</a>.
- Osborn DP. The poor physical health of people with mental illness. West J Med. 2001; 175(5):329–32.
   Epub 2001/11/06. PMID: 11694483.
- Kupferberg A, Bicks L, Hasler G. Social functioning in major depressive disorder. Neurosci Biobehav Rev. 2016; 69:313–32. Epub 2016/07/11. https://doi.org/10.1016/j.neubiorev.2016.07.002 PMID: 27395342.
- Eccleston C, Fisher E, Law E, Bartlett J, Palermo TM. Psychological interventions for parents of children and adolescents with chronic illness. Cochrane Database Syst Rev. 2015;(4):Cd009660. Epub 2015/ 04/16. https://doi.org/10.1002/14651858.CD009660.pub3 PMID: 25874881.
- 27. Bromley J, Hare DJ, Davison K, Emerson E. Mothers supporting children with autistic spectrum disorders: social support, mental health status and satisfaction with services. Autism. 2004; 8(4):409–23. Epub 2004/11/24. https://doi.org/10.1177/1362361304047224 PMID: 15556959.